Title: Successful aging defined by health-related quality of life and its determinants in community-dwelling elders

Authors:

Chia-Ing Li (a6446@mail.cmu.edu.tw)
Chih-Hsueh Lin (d5496@mail.cmu.edu.tw)
Wen-Yuan Lin (wylin@mail.cmu.edu.tw)
Chiu-Shong Liu (d3350@mail.cmu.edu.tw)
Chin-Kai Chang (d13652@mail.cmu.edu.tw)
Nai-Hsin Meng (d6351@mail.cmu.edu.tw)
Yi-Dar Lee (tcli@mail.cmu.edu.tw)
Tsai-Chung Li (tcli@mail.cmu.edu.tw)
Cheng-Chieh Lin (ccclin@mail.cmu.edu.tw)

Version: 2 Date: 18 October 2013

Author's response to reviews:

Dear Editor,

More contextual information in the background section was provided in abstract. Many thanks.

Your sincerely,

Cheng-Chieh Lin